Author's response to reviews

Title: Inferior Alveolar Nerve Injury With Laryngeal Mask Airway-A Case Report

Authors:

Deepak Hanumanthaiah (drdeepakhrh@yahoo.co.in)
Sarmad Masud (docsarmad@hotmail.com)
anil ranganath (dranil_2@rediffmail.com)

Version: 5 Date: 6 November 2010

Author's response to reviews:

1]Author list: changed as per submission order
2]Affiliations: changed
3]References: changed to consecutive numbers in square brackets in text
4]Author contributions: modified
5]Title: all unnecessary capitals removed
6]Competing interests: initials of authors added